Author's response to reviews

Title: Production and characterisation of monoclonal antibodies against RAI3 and its expression in human breast cancer

Authors:

Hannah Jörißen (hannah.joerissen@molbiotech.rwth-aachen.de)
Nuran Bektas (nbektas@ukaachen.de)
Edgar Dahl (edahl@ukaachen.de)
Arndt Hartmann (arndt.hartmann@uk-erlangen.de)
Anette ten Haaf (atenhaaf@ukaachen.de)
Stefano Di Fiore (difiore@molbiotech.rwth-aachen.de)
Hans Kiefer (hans.kiefer@m-fold.com)
Andreas Thess (andreas.thess@curevac.com)
Stefan Barth (stefan.barth@ime.fraunhofer.de)
Torsten Klockenbring (klockenbring@molbiotech.rwth-aachen.de)

Version: 3 Date: 18 June 2009

Author's response to reviews: see over
Dear BioMed Central Editorial Production Team,

We now have addressed the editorial revisions as requested:

- We indicated that all authors have read and approved the final manuscript.
- We provided full reference for reference 25.
- We reduced the volume of figure legend where appropriate.
- We changed figure citations to a consistent manner (Figure).
- We added references to the additional files in the text. Please note that we have changed the order of the files according to their appearance in the text. Therefore, we uploaded renamed figures.
- We deleted the subheading “E-mail addresses” from the title page.
- We rearranged table titles and legends.
- We changed JPEG to JPG for the supplementary figures.

Thank you for accepting our manuscript for publication

Yours sincerely

Prof. Dr. Dr. Stefan Barth